

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Campero, Alejandra; Kaiser, Karen

Working Paper

Access to credit: Awareness and use of formal and informal credit institutions

Working Papers, No. 2013-07

Provided in Cooperation with:

Bank of Mexico, Mexico City

Suggested Citation: Campero, Alejandra; Kaiser, Karen (2013): Access to credit: Awareness and use of formal and informal credit institutions, Working Papers, No. 2013-07, Banco de México, Ciudad de México

This Version is available at: https://hdl.handle.net/10419/83717

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Banco de México Documentos de Investigación

Banco de México Working Papers

 $N^{\circ} 2013-07$

Access to Credit: Awareness and Use of Formal and Informal Credit Institutions

Alejandra Campero

Banco de México

Karen Kaiser
Evercore Partners

June 2013

La serie de Documentos de Investigación del Banco de México divulga resultados preliminares de trabajos de investigación económica realizados en el Banco de México con la finalidad de propiciar el intercambio y debate de ideas. El contenido de los Documentos de Investigación, así como las conclusiones que de ellos se derivan, son responsabilidad exclusiva de los autores y no reflejan necesariamente las del Banco de México.

The Working Papers series of Banco de México disseminates preliminary results of economic research conducted at Banco de México in order to promote the exchange and debate of ideas. The views and conclusions presented in the Working Papers are exclusively of the authors and do not necessarily reflect those of Banco de México.

Access to Credit: Awareness and Use of Formal and Informal Credit Institutions*

Alejandra Campero[†]
Banco de México

Karen Kaiser[‡]
Evercore Partners

Abstract: In this paper we study the determinants of use of formal and informal credit sources. Given that awareness is a necessary step towards use of credit, in order to control for the possible selection bias we decompose the decision to use credit as a two stage decision process in which first, households form their choice set by deciding which type of institutions they want to consider as possible lenders (awareness), and then choose among them (use). Additionally, we allow for correlation between being aware of a specific source of credit and using it. We find evidence that supports the hypothesis that the formal and informal credit markets in Mexico attend different segments of the population. However, our results also show that informal lending sources' characteristics are valued per-se by consumers in certain situations, such as emergencies.

Keywords: Credit demand, consideration set, informal credit, formal credit, Mexico. **JEL Classification**: D1, D14, G2.

Resumen: En este artículo se estudian los determinantes del uso de fuentes formales e informales de crédito. Debido a que el conocimiento de una fuente de crédito es necesario para su uso, para controlar por posible selección descomponemos el proceso de decisión de los hogares de usar fuentes de crédito en dos etapas. En la primera etapa de este proceso los hogares conforman su conjunto de elección al decidir qué tipo de instituciones quieren considerar como posibles prestamistas (conocimiento) y en la segunda, los hogares deciden qué fuente usar dentro de las fuentes que pertenecen a su conjunto de elección (uso). Además, permitimos que haya correlación entre el conocimiento de una fuente de crédito específica y el uso de esa fuente. Se encuentra evidencia que apoya la hipótesis de que el crédito formal y el informal en México atienden a diferentes segmentos de la población. No obstante, los resultados muestran que las características de las fuentes de crédito informales son valoradas per-se por consumidores en ciertas situaciones como por ejemplo, en emergencias.

Palabras Clave: Demanda por crédito, conjuntos de elección, crédito informal, crédito formal, México.

^{*}We want to thank Rainer Schwabe and seminar participants at Banco de México for their comments.

[†]Dirección General de Investigación Económica. Email: acampero@banxico.org.mx.

[‡]Evercore Partners México. Blv. Manuel Ávila Camacho #36 piso 22, Lomas de Chapultepec, 11000 México, D.F. Email: kkaiser@evercore.com.mx.The views expressed herein are those of the authors.

1 Introduction

It has been widely acknowledged that broad financial services have a positive impact on growth and welfare (Claessens, 2005; Demirgüç-Kunt and Levine, 2008; Clarke, Xu, and Zou, 2003; Honohan, 2004; Dehejia and Lleras-Muney, 2007; Levine, 2005). It is in this context that it becomes pertinent to understand the determinants of credit access in Mexico. Moreover, the literature on credit has found that limited access to formal financial services could encourage the development of informal financial institutions which could act as a complement or substitute of the formal sector (Eswaran and Kotwal, 1989; Braverman and Stiglitz, 1989; Kiiza and Pederson, 2002). Thus, it is relevant to understand the demand for credit in a context where formal and informal institutions coexist.¹

Awareness of credit institutions, either formal or informal, is an essential demand-access element of credit. Individuals cannot choose to ask for a loan in an institution that they don't know. It has been recognized that access to financial services must be studied from the supply and demand sides of the market (Claessens, 2002; Claessens, 2006 and Beck and Demirgüç-Kunt, 2008). With respect to the demand side of the market Claessens (2002) explains that individuals are voluntarily excluded of the credit market if they are not aware of the service, do not need the service, or assume rejection. Thus, she suggests that in order to have access to the credit market individuals should decide to be aware of the services. Even though awareness is the first step towards use, not much has been explored about the determinants of awareness of credit sources and their use.

In this manner when we study use of lending sources we face a selection problem. In order to deal with this problem, we study demand for credit as a two stage decision process in which individuals first choose which type of institutions they want to consider as possible lenders and then choose among them. Using household data for Mexico we estimate the first stage using a model of credit use with consideration set formation (Andrews and Srinivasan, 1995). The estimation strategy uses all the information about household's knowledge on all possible credit sources rather than focusing on the knowledge of a particular credit institution. Additionally, it allows us to exploit information on households that are excluded from the formal, informal or both credit markets because they are not aware of the institutions, rather than focusing only on the households that are already participating in the credit market. In the second stage estimation we model the decision process as a multinomial logit taking as given the household's choice set formed in the first stage. To deal with the possible

¹Formal lenders are institutions regulated by the government and the Central Bank whereas informal lenders operate beyond the regulatory framework of the financial system (Zeller, 1994). As formal institutions we study banks, savings banks, and government loans. Informal insitutions considered in the study are: money lenders, pawnshops, and family or friends.

selection bias, we allow for correlation between awareness of a specific source in the first stage and its use in the second stage (Green, 2006; Terza, 2002).

The estimation in this paper allows us to disentangle the effect of household and locality characteristics on awareness and use of different types of credit. The results highlight the importance of knowing the specific channel (use or awareness) through which different variables affect access to formal and informal credit. It also allows us to differentiate those households that are aware of credit sources but decide not to participate from those that are not aware of credit sources.

In general, our results suggests that the formal and informal credit sectors are attending different types of households. However, we also find evidence that the informal sector plays a complementary role to the formal sector. There is a positive relationship between income and use of formal institutions such as banks and savings banks but a negative relationship between income and the use of informal sources such as money lenders and friends or family. With respect to formal credit institutions we find that schooling, age, and type of employment affect awareness of these institutions. With respect to the determinants of informal credit demand, we find that household size has a positive effect on awareness and use of informal credit institutions. Women use more pawnshops as lending sources and education has a negative relationship with the use of this source.

Moreover there is some evidence that suggests that the characteristics of the institutions that conform the informal sector such as their flexibility and rapid access to loans, are valuable to households (Pearlman, 2010; Beck and de la Torre, 2007). Such is the case of family and friends as lending institutions. We find that the awareness and use of family and friends as credit lenders increase when households suffer a negative income shock. This proposes that some characteristics of this credit institution make it a better choice than formal institutions when the household suffers a negative shock. This points out to the existence of some complementarity between the formal and informal sectors in Mexico.

The rest of the paper is organized as follows: the next section gives a description of the financial sector in Mexico, Section 3 provides a literature review; Section 4 describes the methodology used in the estimation; Section 5 presents the data and summary statistics; Section 6 presents the results; and Section 7 concludes.

2 Formal and informal credit institutions in Mexico

This paper studies awareness and use of both formal and informal credit sources in Mexico. Following the definition of formal lenders being institutions regulated by the government who operate inside the regulatory framework of the financial system, the formal sources addressed in our study are: banks, savings banks, and government credit programs; while, informal sources are: money lenders, pawnshops and family or friends.

Banks are formal financial institutions that are primarily engaged in accepting deposits from the public for active credit operations. Banks can be either commercial or development banks. Commercial banks are privately owned while development banks are publicly managed and target their credit to specific activities.

Savings bank and cooperatives are both regulated financial institutions and they are considered to be Peoples Financial Sector institutions by the National Banking and Securities Commission (CNBV). For this reason savings banks and cooperatives are typically located in low income municipalities. They mainly offer savings and credit services but can also offer other services such as payments from government social programs, recipient of remittances, payment of electricity or water bills, and insurances. The difference between savings banks and cooperatives is that while cooperatives are nonprofit institutions that provide services only to their members, savings banks offer services to the general public.

Government credit in this study mainly considers two programs One's word credit (Crédito a Palabra) and Mexican Institute of Social Security's credit (IMSS) or Institute of Social Security and Services for Government Workers' credit (ISSSTE). One's word credits are funds that are granted to low-income farmers that have no more than 20 hectares of land under cultivation during the rainy season and that are not eligible for bank credit. The payment of these credits stays in the community and is used for work that benefits the community or to promote savings banks. IMSS and ISSSTE are in charge of the provision of public health systems. ISSSTE covers federal government workers only and IMSS covers formal private workers. They offer registered individuals credits such as house funding or funeral support for the death of family members. Additionally, IMSS gives out loans to employees from their own pension funds and they pay them through their monthly salary. ISSSTE gives out personal loans according to the employees' salary and the time they have been working for the federal government.

A money lender is considered to be any group or person that gives out loans. Money lenders can get resources form personal surpluses or borrow from other credit institutions (Toby, 1991). The literature considers that they can lower information costs and that they present the highest interest rates (Varghese, 2005; Hernández-Trillo, Pagán and Paxton,

2002). According to Hernández-Trillo, Pagán and Paxton (2002) in Mexio money lenders offer loans for an average term of 4 to 9 months, which is considered medium term investment, while friends and family give out short term loans and banks long term credit. However, there is not much information on use or availability of money lenders.

According to the Mexican Economic Census data, pawnshops are a growing business in Mexico, even when they have monthly interest rates that go from 10 to 23%. The number of pawnshops went from 1,589 to 5,948 from 2004 to 2009. Moreover, in 2010 according to Federal Consumer Protection (PROFECO) that is in charge of the regulation of pawnshops, 90% of pawnshops are private or commercial and have an average annual cost of 257% of the borrowed amount; the other 10% of the pawnshops are institutions of private assistance and have an average annual cost of 120%. The average loan in pawnshops varies between the 25% and 45% of the value of the pledge pawned. PROFECO also reports that 80% of the pawnshops users are housewives and the other 20% are in their majority merchants, students, unemployed and retired people.

Regarding credit from friends and famlily, according to National Survey of Microenterprises (ENAMIN) loans from this source are also the most frequently used at the time a firm starts up in Mexico.

3 Literature Review

Previous literature has provided evidence of the effect that financial services have on economic growth and poverty alleviation (Claessens, 2005; Demirgüç-Kunt, and Levine, 2008; Clarke, Xu, and Zou, 2003; Honohan, 2004; Dehejia and Lleras-Muney, 2007; Levine, 2005; Peachey and Roe, 2004). Beck, Demirguc-Kunt and Martinez (2008) find that both outreach and depth indicators of financial systems are positively correlated with economic development, quality of institutions and development of physical infrastructure. Moreover, access to finance for large parts of the population has been recognized to be important to expand opportunities, for driving democracy and market economy, and enhance technological progress (Rajan and Zingales, 1998; King and Levine, 1993). Furthermore, access to finance has been compared with access to basic needs such as water, education, and health services (Peachey and Roe, 2004).

The literature on access recognizes that access to financial services is not only a matter of availability of institutions but an interaction between access-supply and access-demand factors (Claessens, 2006; Beck and de la Torre, 2007; Beck, Demirguc-Kunt and Martinez, 2008). With respect to the supply side of the market (supply-access), access has been widely identified with availability of services (Claessens, 2005). From the demand side (demand-access),

Claessens (2002) classifies types of access in three groups, two groups that are excluded from the market and one that is not. The first group is called involuntary exclusion and it is composed by people that have been rejected due to their high risk profile, discrimination, price, product, income, or respondent features. The second group, voluntary exclusion group, includes those that do not need the service, those that are not aware of the service, and those that assume rejection. Finally, the third group is composed by those that are consumers of financial services. Furthermore, this literature recognizes that it is important to distinguish between access and use when discussing financial sector outreach and finds that the actual demand for credit can be lower due to lack of access because of reasons such as financial illiteracy or lack of awareness (Beck and de la Torre, 2007).

Another strand of literature that is relevant for our work is the one related to the role of the informal financial system in demand for credit. It has been explained that the existence of scale economies and network externalities makes individuals living in small communities or individuals with the need for small credit transactions unprofitable for formal financial service providers (Peachey and Roe, 2004). Because of the limited access to formal financial services, it has been argued that the informal sector faces the residual demand from the formal sector (Eswaran and Kotwal, 1989; Braverman and Stiglitz, 1989; Kiiza and Pederson, 2002). In this sense, it has been said that the poor rarely have access to voluntary deposit or credit services offered by formal institutions so they are obliged to save or ask for a credit in an informal source, which often has higher costs, risks, illiquid and indivisibility (Beck and de la Torre, 2007). On the other hand, it has been explained that formal and informal credit sources interact horizontally especially when low-cost credit is available through informal sources (Bell 1990; Kochar 1997; Conning and Udry, 2007; Guirkinger, 2008).

In particular, previous literature has considered friends and family loans to be a way of risk sharing that is beneficial for a group of people, specially in developing countries like Mexico and among individuals that are exposed to shocks in their income. Furthermore, credit between friends and family are considered to be effective because of peer monitoring (Besley, 1995). Pearlman (2010) finds that in Peru and Ecuador family and friends are the main source of funds when dealing with an enterprise or family emergency. The author explains that this is mainly because flexibility may be very important to poor borrowers who face high levels of risk and limited means of managing it. Large shocks such as robbery, bribes, extrusion and natural disasters limit the ability to meet rigid repayment schedules usually adopted by formal credit sources. For these reasons, in this paper we use data on awareness and use of formal and informal services which allows us to characterize the determinants of access of both types of sources in Mexico.

With respect to use of credit, access to formal and informal financial services has been

studied so far as a decision process in which the individual or household decides whether to use or not a specific service. In this sense, Besley and Levenson (1996) and Anderson and Baland (2002) study participation in Roscas in Taiwan and Kenya respectively using a one-stage probit model. The use of financial services from the demand side has also been studied through the use of multinomial models. By using multinomial probit models Bendig, Giesbert, and Stainer (2009) simultaneously estimate the demand (use) for savings products, loans, and insurances in Ghana. Wydick, Hayes and Kempf (2011), using a multinomial logit model, measure the extent to which social networks determine use of different sources of credit in Guatemala. Finally, Tang, Guan and Jin (2010) use a multinomial product model to study the demand for formal and informal credit in China.

Use of financial services has also been studied through two-stage models. Zeller (1994) uses univariate probit models to see how individual and household's characteristics and events might affect the probability of an individual asking for credit and being granted such credit. To see the difference between formal and informal sources of credit, the author estimates the model separately for the formal and informal sectors. In order to study whether formal credit rationing in rural India exists, Pal (2002) first estimates the probability that a household demands informal or formal credit using a multinomial logit model and then, estimates the demand on informal loans conditional on having one formal loan. However, none of these studies account for the fact that no participation/use could be due to lack of awareness of the financial institutions that are studied.

Our paper follows more closely the literature that considers the selection that exists derived from the awareness of a specific source when studying use of financial services. Okaten and Osil (2004) estimate the impact of social networks on access to credit in Indonesia with a non-simultaneous, three-stage selection model. The authors explain that the survey used presents a selection bias due to the fact that only individuals that report knowing a source of credit are asked whether they borrowed or not. In order to correct this bias, the first stage in this paper presents an equation for selection based on whether the individual knows any source of credit or not. They find that social networks are important in knowing a place to borrow, as well as for loan approval. Differently, in our paper we consider each of the lending sources a household is aware of (we study 6 different sources, 3 formal and 3 informal). Thus in the second stage of our model we use a multinomial logit while Okaten and Osil (2004) use a probit model.

Finally, our estimation is based on papers that explain how to model two-stage multinomial logits. In this sense we used Andrews and Srinivasan's (1995) consideration set formation model. We also considered literature on instrumental variables in multinomial logits such as Terza (2002), which studies the effect of alcohol abuse on employment conditions

and addresses endogenity in a multinomial logit model. Additionally, we used Green's (2006) multinomial logit model with sample selection. Lastly, we based the computational estimation of our model on Haan and Uhlenforff (2006) suggested routine for multinomial logit models with unobserved heterogeneity using maximum simulated likelihood.

What makes our model different form all others is that, first, we study the demand of credit from various institutions simultaneously through a multinomial logit, second, we correct for selectivity due to awareness of a source, and third, when we estimate the household decision to use a specific source of credit we allowed each household to choose only from the sources it is aware off.

4 Estimation

According to previous literature, choosing which alternatives to consider when making a decision is relevant because a choice might be made when alternatives are not physically present, because consumers must know what they are looking for or because households lack the motivation to locate and examine multiple alternatives (Andrews and Srinivasan, 1995). Moreover, consumers create consideration sets with specific characteristics, they prefer to create a consideration set from alternatives that are easier to compare or that have a high likelihood of containing their optimal alternative. Consumers often use their past experiences to narrow down their awareness set, this is called screening criteria and it is used when consumers face a large set of alternatives in order to reduce the number of alternatives they will choose among. Screening criteria can rely on familiarity of the alternative or memory accessibility (Chakravarti and Janiszewski, 2003). Additionally, as it has been mentioned earlier, the literature on access to credit considers unawareness of credit sources as voluntary exclusion which suggests that individuals decide to be aware or not of lending institutions (Claessens, 2006).

Given that we can only observe the decision to ask for credit from those individuals that are aware of a specific source and that individuals choose which sources to know, there is a selection bias on the sample that uses financial services. In order to deal with this problem we use the two stage model developed by Andrews and Srinivasan (1995) correcting for selection as suggested by Greene (2006) and Terza (2002).² In the first stage households decide which institutions they will consider when deciding to ask for a loan, i.e. each household forms its consideration set of lending institutions. In the second stage, decision makers decide which

²Andrews and Srinivasa (1995) developed the model to study the demand with consideration set formation for scanner data. Consideration set formation is the process in which the individuals decide which options they want to consider when making a decision.

institution among the ones in their consideration set maximizes their utility. We allow for correlation between awareness of a source and the use of that specific source in order to account for the endogeneity that might exist between being aware and use of a source.

In our model there are six different institutions that provide loans: $K = \{\text{banks, savings banks, government, money lenders, pawnshops, family and friends}\}$. We observe the use/non use of a specific institution only if the household knows the source. A household (j) knows an institution (k) if $z_k^{j*} = 1$ $(z_k^j > 0)$, where

$$z_k^j = \alpha X_{jk} + \gamma W_{jk} + \eta_{ik}.$$

Where $\eta_{jk} \sim N[0,1]$, X_{jk} referes to household characteristics and W_{jk} is a set of instruments for being aware of a source and that are not directly related with use of a lending source. Thus, for every household we have a consideration set $C_j \subset P$, determined by $\{z_k^{j*}\}_{k=1}^6$, and P is the set of all possible subsets of set K (lending institutions) excluding the empty set. We assume that in addition to the institutions that the household decides to consider (household's j consideration set), the household's choice set $(S_j \subset \Omega)$ also includes the option not asking for a loan, i.e. $S_j = C_j \cup \{not \, asking\}$. Hence, the set of possible choice sets is $\Omega = \bigcup_{C_j \in P} \{C_j\} \cup \{not \, asking\}$.

Therefore, in the second stage the household decides to apply for a loan or not, taking as given it's choice set S_i . The household maximizes its utility and solves:

$$\max\{u_{jk}\}_{k\in S_j}$$

where $u_{jk} = X_{jk}\gamma + \sigma_k v_j + \varepsilon_{jk}$ and $\varepsilon_{jk} \sim GEV$ (generalized extreme value distribution), $v_j \sim N[0,1]$ is a random parameter that captures the unobserved factors through which sample selection operates. Following Greene (2006) the probability that household j chooses option $k \in S_j$ follows the multinomial logit model with sample selection:

$$p(k|S_j, X_{jk}, \upsilon_j) = \begin{cases} \frac{e^{\left(X_{jk}\gamma + \sigma_k \upsilon_j\right)}}{\sum_{n \in S_j} e^{\left(X_{jn}\gamma + \sigma_n \upsilon_j\right)}} & \text{if } k \in S_j \\ 0 & \text{if } k \notin S_j \end{cases}$$

$$(4.1)$$

Note that in our case the data are characteristics of the individual rather than attributes of the choices, then the coefficients are renormalized. In the second stage of the estimation (use) not asking for a loan will be used as the baseline category and then all the coefficients should be interpreted with respect to it.

In order to correct for the possible endogeneity between the first stage decisions and the second stage, we allow for correlation between the error terms of the first and second stage: $(\eta_{jk}, v_j) \sim N[(0,1), (1, \rho_k, 1)]$. By joint normality $f(\eta_{jk}|v_j) = N[\rho v_j, (1-\rho^2)]$. The parameters σ_k , in equation (4.1), and ρ will correct for the endogeneity between the sencond and first stage.

In order to construct the maximum likelihood p(k) we need the probability of choosing institution k given that the household has choice set $S_j : p(k|S_j)$, and the probability of being aware of the institutions belonging to the set $S_j : p(S_j)$. The latter probability is computed as:

$$p(S_{j}|X_{jk}, W_{jk}, v_{j}) = \prod_{n \in S_{j}} p(z_{n}^{j*} > 0) \prod_{n \notin S_{j}} \left(1 - p(z_{n}^{j*} > 0)\right)$$

$$= \left[\prod_{n \in S_{j}} \Phi\left(\frac{\alpha X_{jn} + \gamma W_{jn} + \rho_{n} v_{j}}{\sqrt{1 - \rho_{n}^{2}}}\right) \prod_{n \notin S_{j}} \Phi\left(-\frac{\alpha X_{jn} + \gamma W_{jn} + \rho_{n} v_{j}}{\sqrt{1 - \rho_{n}^{2}}}\right) \right]$$

Note however, that v_j is not known, is a random variable that captures the unobserved heterogenety. Thus in order to compute the maximumlikelihood function we need to integrate over the distribution of v_j . We assume that household heterogeneity v_j is the same among households and is independent across households. Then, the probability of observing the demand for credit in institution k is:

$$L = p(k) = \prod_{j=0}^{J} \int_{-\infty}^{\infty} \left(\sum_{S_j \subset \Omega} p(k|S_j, X_{jk}, W_{jk}) p(S_j|X_{jk}, W_{jk}) \right) f(v_j) dv_j$$
 (4.2)

We estimate equation 4.2 by simulation in STATA.³

$$\ln L = \sum_{j=1}^{J} \ln \left[\frac{1}{R} \sum_{r=1}^{R} \left(\sum_{S_j \subset \Omega} p(k|S_j, X_{jk}, W_{jk}, r) p(S_j|X_{jk}, W_{jk}, r) \right) \right]$$

The simulation draws R values (r) from the unobserved heterogeneity (v_j) distribution, for each draw the maximumlikelihood is estimated and then the average is computed. Instead of using random draws we follow Green (2006) and Hann and Uhlendorff (2006) and base the simulation using Halton sequences. It has been shown that Halton draws provide a wider coverage of the range of integration (Hann and Uhlendorff, 2006).

³The code is avaliable upon request to the authors.

5 Data and summary statistics

We use data from The Mexican Family Life Survey (MxFLS) for the year 2002.⁴ MxFLS is a longitudinal household survey that collected information from a representative sample of approximately 8,400 households or 35,000 household members in 150 localities throughout Mexico. The survey includes information on household members, households, and localities. The survey covers household member characteristics like educational attainment, labor market participation, and income; household characteristics such as assets, expenditure, and credit availability; and locality characteristics such as population, credit opportunities, and infrastructure. MxFLS is representative at the national, urban-rural, and regional level. Data for 2005 has also been released by MxFLS, however expansion weights are under construction and not available. Due to the fact that the design of the survey sample is very complex (three-staged, by conglomerates and stratified) the use of expansion weights is essential.

In the present study we use the credit survey in MxFLS answered by the members of a household that are 15 years old or older. In this survey the interviewer first ask if the household member knows a credit source.⁵ If the answer is "no" then they don't ask any more credit questions; if the answer is "yes" then they name different options from which household members have to specify the ones they know. If the household member does know at least one source of credit they proceed to ask if in the last 12 months they have borrowed from any of the sources they know and to specify from which of them have they borrowed. The fact that credit questions are only asked to household members that are aware of a source of credit presents a sample selection bias when studying use of credit.

The survey provides information about eleven different types of credit sources: bank, savings bank or cooperatives, *Crédito a Palabra* program, IMSS or ISSSTE credit (social security institutions), other government credit, money lender, pawnshop, work, family, friends, and other. We grouped friends and family and the three types of government credit (*Crédito a Palabra* program, IMSS or ISSSTE credit and other government credit), in order to have more observations in those groups.⁶ Due to the lack of detail on what sort or credits are work credits and other they were not considered in this study. We were left with six different sources of credit: bank, savings bank, government credit, money lender, pawnshop and family or friend.

⁴MxFLS only exists for the years 2002 and 2005.

⁵ "Do you know any person or place where you can borrow or ask for a credit?" The answer can either be "yes" or "no" and that is why our awareness variable takes values 1 and 0 only.

⁶Other sources of credit were not grouped because they have unique characteristics and have different effects on outcome variables when comparing them to the rest of the credit sources.

In our analysis a unit of observation is a household. This is because the survey provides more information at a household level than at a household member level. Additionally, it is not strange to think that awareness of a source of credit is easily shared among the household members and that income is shared within the household members, therefore if a household member needs a loan other household member could act as an intermediary. The fact that households are the unit of observation means that a specific source of credit is part of a household's consideration set if there is at least one member in that household that knows that source of credit. Hence, the choice set of a household is the union of all the different sources of credit known by their members.

Similarly, a household is considered to have asked for a credit from a bank, savings bank, government credit, money lender, pawnshop, and family or friend if at least one of its members asked for a credit in that source. Since an important assumption in multinomial logit models is that only one alternative is chosen, for the households that have loans in more than one source of credit, we assign the source in which the borrowed amount was the largest.⁸

Table 1 provides a description of the sources of credit that households are aware of. Panel A shows the number of credit sources that households in the sample know, which is the cardinality of their consideration set. As it can be observed, 47% of the households in the sample do not know any of the six sources of credit considered in this study, therefore their consideration set is empty. The large amount of households that report not knowing a source of credit can be due to a measurement error in the survey. This can be caused by the way household members interpret this question. Knowing a source of credit has a wide range of interpretations, from being aware of its existence to having the knowledge of payment conditions and interest rates. However, it would be difficult to dimension this measurement error due to the impossibility of comparing the data with other surveys because of the particularity of the characteristics and questions in the MxFLS. Around 35% of the households are aware of only one source of credit while 17% of the households in the sample are aware of two or more different types of credit. Moreover, only 0.1% of the households in the sample is aware of all the credit sources considered in this study. Panel A also provides a description by source of credit. It can be seen that the source that households are more aware of is family and friends, followed by bank, and the one that they are least aware of is government credit. This also holds for households that are aware only of a single credit source. Out of the households that are aware of more than one credit source a large number

⁷Data at household member level has a lot missing values and therefore very few observations. Due to the large number of parameters that are estimated in the model, it needs more observations to converge. In order to have more observations data was aggregated by household.

⁸Only 1.97% of the households in the sample report having a loan in more than one source of credit.

of them have family or friends as part as their choice set. Around 38% of the households that know a family or friend from which they could get a credit also know other sources of credit, making family and friends the largest alternative source of credit known. For example, 73% of the households that know a bank in which to ask for a loan know other sources of credit. 48% know a family and friend and only 25% know an other source.

Table 1, Panel B presents the percentage of households in the sample that know and demand each type of credit. Out of the 53% of households that do know a source of credit only 16% applied for a credit. Family and friends is the most commonly known and used source of credit, 35.7% of households know a family member or friend who could lend them money and 10.1% of the households are actually in debt with a family member or a friend. From Panel B it can be seen that bank occupies the second position in sources of credit known but only 0.7% of households actually demand a credit from a bank. Government credit lends only to 0.6% of the households.

To control for household characteristics we use the household size, which is the number of household members; the household head's sex (dummy variable where 1 is male and 0 female) and age; average schooling of the head and spouse in a household; the log of the household's annual income, where a household's annual income is the sum of the annual income of all of its members. Controls such as negative shocks and employment condition are built from other variables in the survey. A household is considered to have a negative shock to its income if the household has experienced one or more of the following events in the past 5 years: death of a household member, illness or serious accident that required hospitalization, unemployment or business failure, house or business lost due to an earthquake, flood or other natural disaster, total loss of crops, robbery, or death of production animals. We use two variables on the employment condition of the head of the household as controls, these variables are whether or not the household head is formally or informally employed. To define formal employment we consider two cases: first, an employee is considered to be formal when she receives social security, private medical insurance provided by the employer or pension found; second, a self-employed worker is considered to be formally employed if she receives social security. A worker is considered to be informally employed if she works but is not a formal worker.

Locality controls were built using the variable to identify localities in the survey.⁹ The average income in a locality is the average income of the households that report their income in a locality.¹⁰ Table 2 shows the means of the variables in the sample at a household level,

⁹Even though this survey is not representative at a locality level, due to the lack of locality identifier, locality controls were constructed using the survey and not Census data.

¹⁰In order to control for geographic barriers we also tried using dirt roads, post office, and bus terminal in locality. Additionally, we tried using income and schooling variables from the Mexican Census (a representative survey) at municipality level. Results didn't change.

therefore the mean of the variables bank in locality, savings bank in locality, government credit in locality, pawnshop in locality and money lender in locality are the percentage of households that live in a locality that has that source of credit.

In order to identify availability of credit sources in the locality we create a set of variables that take value one if there is a branch of that type of credit institution in the locality and zero otherwise. Bank in locality, savings bank in locality, government credit in locality and pawnshop in locality are variables built from the *Communities* questionnaire in the survey. Money lender in locality was not used due to the large proportion of missing values this variable has. A locality is considered to have access to government credit if the members of the locality are beneficiaries of programs such as government credits (Créditos Gubernamentales) or housing subsidies (Ahorro, Subsidio y Crédito para la Vivienda Progresiva, VivAh). We use availability variables in order to control for transportation costs.

Table 3 presents the percentage of localities by the number of sources of credit they have and the percentage of the households in the sample that live in these localities. It can be observed that 39.1% of the localities in the sample don't have any source of credit and 31.8% have only one source of credit. We can also observe that only 4.6% of all localities in the sample have the four different credit sources. The fact that there are so many localities without credit sources or with a single type of credit source, and so few localities with all types of credit sources means that the members of a household must travel to other localities in order to use a specific type of credit.

In the sample only 41% of the households live in localities where there are banks and 39% live in localities where there are saving banks. In the sample, government credit is the sources of credit that most localities have. The percentage of localities that have government credit as its only source of credit is around 27.2% of all localities in the sample and 64% of the households live in a locality where government credit is offered, most of these localities are rural and have an average income below the mean. Differently, pawnshops are the source least present in localities in the sample and only 25% of the households in the sample live in localities where there are pawnshops. Pawnshops are only situated in localities where there is at least another different source of credit.

In order to avoid identification problems selection models include independent variables in the selection equation that are excluded from the main equation. These are instrumental variables and they have to fulfill two conditions. First, that instruments are strongly correlated with awareness and second, that they are uncorrelated with use of credit. In this study, the variables used for awareness (selection equation) that are excluded from use are: previous serious accident or health problem, previous natural disaster in the locality, public

transportation in the locality, and change in infrastructure in the locality. 11 A household is consider to have had a previous serious accident or health problem if they report that they had one in the past but they are no longer affected by it. ¹² A locality is considered to have a previous natural disaster if there was a flood, earthquake, landslide, fire, hurricane, draught, plague, frost, or hailstorm before 2000. The existence of public transportation inside the locality is directly reported in the Communities questionnaire. A change in infrastructure in the locality is considered to be the construction of roads, introduction of drainage, introduction of clean water systems, telephone, construction of health services, and construction of new schools. Summary statistics of these variables are also presented in Table 2. These variables fulfill the two conditions mentioned above. First, we find that they are strongly correlated with awareness by conducting a Wald test where the null hypothesis is that all the parameters of the instruments in each of the awareness equations are jointly equal to zero. The null hypothesis is rejected at the 5% level for awareness of bank and money lender, and at the 15% level for awareness of savings bank, government credit, pawnshops and family and friend. They also fulfill the second condition. Previous incidents that happened a long time ago and do not affect households' income in the present are correlated with awareness of sources because they might have needed to borrow money in the past and aren't correlated with use because they are no longer affected by these shocks. Public transport is correlated with awareness of credit sources because it allows household members to cover longer distances and get to know more sources, and it does not explain use of credit. Finally, infrastructure not only increases communication inside the locality but also might become more attractive for credit sources to open a branch in that locality. Both situations increase information about sources of credit without affecting its use.

6 Results

In Table 4 we present the results of the model with consideration set formation. Panel A shows the results for awareness and Panel B the results for use conditional on awareness for each source allowing for correlation between the errors in the first and second stage. Note that awareness and use of the six sources is computed simultaneously. In order to control for possible error correlation within localities we compute the standard errors using clusters

¹¹ Following Terza (2002) we use variables at locality level to instrument for variables at household level.

¹²This variable is different from negative shock, because negative shock to the income caused by health problem and accident that required hospitalization is in the previous 5 years and can still be affecting the household. Differently, previous accident or health problem refers to a problem that was faced in the past and is no longer affecting the household's income.

at this level. 13

Results in Table 4 Panel B are estimated using the information on the choice set of each household and they present the effect of each variable on conditional use corrected for selection. In order to interpret the results it is important to note that the first stage presented in Panel A (awareness) is modeled using a normal distribution, therefore coefficients can be interpreted as the change in the predicted probability that the source of credit is part of the households' choice set or the probability that the households are aware of it, when there is a change in one of the independent variables holding everything else constant. Table 5 Panel A presents the marginal effects at the mean. The base category in the estimation in Panel B (use) is not asking for a loan, so all the results in this column should be interpreted with respect to this variable. The magnitude of the coefficients in the second stage (use) can be interpreted as coefficients of a standard multinomial logit model using odds ratios. The odds ratios for the second stage are presented in Table 5 Panel B.

From Table 4 we can see the effect that different variables have on the awareness and use of different sources of credit. We find evidence that suggest that the formal and informal sectors attend different segments of the population and that the informal sector seems to have a complementary role derived from its characteristics such as flexibility and rapid lending procedures.

Our estimation provides information on the segmentation of the credit market when analyzing how income affects demand of different credit sources. Results show that there are two different effects of income. The first effect is positive and is through awareness of credit sources, the probability of being aware of credit sources, either formal or informal, is increasing in income. An increase of 1% of income rises the probability of being aware of bank, savings bank, government credit, and family and friends by 0.04, 0.03, 0.003, and 0.02 percentage points respectively, in comparison with the probability of households with the mean income. However, the second effect is reflected in use and is different for formal and informal sources. Income has a positive effect on using formal credit and a negative effect on the use of informal credit conditional on awareness. In increase in 1% of annual income increases the odds of asking for a credit in a bank by 2.05 times and decreases the odds of asking for a credit from family and friends 0.2 times. This provides evidence of what has been mentioned in past literature about income having a positive impact on access to credit from formal institutions while some sources of informal credit attend the residual demand which includes households with lower income (Beck and de la Torre, 2007; Eswaran

¹³When estimating the model in Stata through maximum likelihood there exists the option vce(clusters id loc), where id loc is the variable that has the locality ID.

¹⁴The interpretation of these coefficients is the same as the interpretation of a probit regression. Awareness of each source of credit is modeled as a probit.

and Kotwal, 1989; Braverman and Stiglitz, 1989; Kiiza and Pederson, 2002). As Peachey and Roe (2004) explain, this could be happening because lower income households ask for small credit transactions unprofitable for formal financial service providers or because poor households have a lack of income stability or collateral. In addition, we find that average income in locality has no effect on awareness but conditional on awareness localities with higher average income use less credit form both formal and informal sources.

Age and schooling affect the demand-access of formal credit institutions. Age and age squared have a positive effect on being aware of formal sources of credit. A household with a household head that is one year older than the mean has a probability of being aware of bank credits 0.002 percentage points greater than a household with a household head that is 46.2 years old (age mean). However, age is negatively correlated with using formal sources. An increase of one year would decrease the odds of using a credit from a bank 0.13 times. Older people are more likely to be aware of these sources, but out of the ones that are aware of them the younger ones are the ones that use them. Schooling is positively correlated with being aware of formal sources, yet, when looking at use, we observe that schooling is no longer significant in explaining use of banks and it is negatively correlated with using savings banks. On one hand, one year of schooling above the mean increases a household's probability of being aware of banks by 0.009 percentage points with respect to those households with average schooling. On the other hand one more year of schooling decreases the odds of using savings banks by 0.14 times. Moreover, we find that higher average schooling in a locality has no effect on awareness but it affects use of formal credit positively.

Employment condition is also an important determinant of the demand of formal sources of credit. A variable that affects banks, in particular, is being formally employed. This variable has no effect on knowing a bank but has a positive effect on using it. For formally employed head of households, the odds of using credit form a bank are 5.71 times as large as the odds for informally employed or unemployed of using such credit. This could be explained because having a formal job might make it easier for the household members to show that they have a stable job and income which is necessary to get a loan from a bank. Awareness of government credit is negatively affected if the household is informally employed. However, use of government credit is negatively affected if the household head has a formal job.

Additionally, we find evidence that suggest that availability of formal institutions affects demand-access. The availability of banks and savings banks has a small but positive impact on households being aware of that particular source. However, it should be noted that the availability of any type of credit source has no effect on use of that particular source. These

results suggest that having access to a source of credit in the locality doesn't have an effect on a household decision on whether to ask for a credit or not conditional on knowing that source of credit. Furthermore, the results show that availability of formal sources has a negative effect on the use of money lender, which provides evidence on the fact that substitution between formal and informal sources of credit exists.

With respect to the informal market a determinant of demand-access is household size. This variable has a positive effect on knowing and using informal sources of credit. An increase of one family member above the mean increases the probability of a household of being aware of money lender, pawnshop and family or friends by 0.006, 0.007, and 0.02 percentage points respectively in relation to households that have the average number of members. An additional member in a household increases the odds of using money lender credits 1.22 times and pawnshop credits 1.30 times. The positive effect on use could be explained by assuming that the household head or spouse have more members to support, they need more resources in their daily expenses and therefore they get loans form informal sources since these sources can lend smaller amounts in little time.

When analyzing pawnshops we notice that this source of credit has some interesting characteristics. Male and female household heads are both equally likely to be aware of pawnshops; however when it comes to using pawnshops women are more likely to use them than men. The male-female odds ratio of using pawnshop credit is 0.75 to one. Additionally, schooling is negatively correlated with use of pawnshops. An increase in a year of schooling decreases the odds of using pawnshops by 0.10 times.

Moreover, our results suggest that the informal sector also acts as a complement of the formal sector. The informal sector, and more specifically credit from family and friends, seems to be an important source of credit to face unexpected events. This can be seen in the positive effect that the variable negative shock to the income has on awareness and on use of family and friends as a source of credit. The fact that a household has faced a negative shock increases the probability of being aware of family and friends and the odds of using credit from family and friends by 1.73 times with respect to those households that have not faced a negative shock to their income. This is consistent with the literature that explains that family and friends networks are used as a way of sharing risk and as funds when dealing with an enterprise or family emergency due to its flexibility and rapid lending procedures (Pearlman, 2010). Negative shock also has a positive effect on knowing government credit; however it is not significant in explaining its use.

We find evidence of selection supported by the significance of ρ in all sources and σ in all except bank. Moreover, we conducted a Wald test where the null hypothesis is that the σ parameters of all sources are jointly equal to zero. The test statistic is equal to 61.88 (p-

value is 0.00), hence the null hypothesis is rejected at the 0.1% level. This provides further evidence on the need of correcting for selection due to awareness when studying use of credit.

7 Robustness

In order to show the relevance of estimating a model of consideration set formation correcting for selection we also present the results of an unconditional multinomial logit model and a consideration set formation model without selection. Table 6 presents the unconditional multinomial logit model where it is assumed that all households know all the credit institutions and choose the type of credit that maximizes its utility. Comparing our results with those of the multinomial logit model allows us to establish more clearly the mechanism, awareness or use, through which variables affects demand for credit. Multinomial logit coefficients sometimes seem to be capturing the awareness effects and sometimes the use effect. This estimation does not capture some important effects that impact demand through awareness and use. For example, the multinomial logit fails to capture the effect of sex when using pawnshops, the impact of household income on use of some informal sources, and the effect that schooling has in formal sources.

Table 7 presents the consideration set formation model without selection. This model fails to capture various effects such as the effect that household size has on being aware of some informal sources of credit, awareness of government credit due to negative shocks, and the effect that sex has when using pawnshops. On the other hand, this model has effects that disappear when correcting for selection such as the effect that sex has on using money lender and savings bank. Moreover, coefficients in this model are under or over estimated. The difference in these results emphasizes the importance of having a two stage model with consideration set formation correcting for selection.

In order to gain statistical efficiency and check our results we grouped the sources of credit in formal and informal. Results are presented in Table 8. We find that some of the general results presented still hold, however, other results disappear because of contrary effects that some variables have within formal or informal sources. Household size still affects awareness of informal sources but no longer its use, due to the fact that it was only significant for two of the four informal sources. We still get that male household heads are more likely to be aware of informal sources and that schooling and formal employment variables have a positive effect on awareness of formal sources. Grouping the sources can be misleading. For example, we find that income affects awareness only of formal sources and it would seem like income doesn't have an effect on the use of any type of credit source. Also, when looking at negative shock variable, we observe that it has a positive effect on knowing both formal and

informal sources and that it has no effect on the use of neither. These results would lead to conclude that informal and formal sources are acting similarly when facing a negative shock when in Table 4 we can see that they aren't.

8 Conclusions

In this paper we estimate the effect that different variables have on awareness and use of formal and informal credit sources in Mexico using a model of consideration set formation and correcting for selection. We assume that the first step for using a specific credit source is being aware of it, therefore it is important to consider the selection problem when studying use of credit. Furthermore, we include the fact that different households are aware of different sources of credit through the formation of households' consideration sets in the estimation.

We find evidence that suggest that the formal and informal sectors attend different segments of the population and that the informal sector seems to have a complementary role derived from its characteristics such as flexibility and rapid lending procedures.

The fact that formal and informal sectors attend different segments of the population can be seen through the impact that some variables have on awareness and use of formal and informal sources. Income has a positive effect on awareness of both formal and informal sources; however, it has different effects in their use, income has a positive effect on the use of formal sources and a negative effect on the use of informal sources. Awareness and use of formal and informal sources are also differently affected by schooling. On one hand, households with higher average schooling are more likely to be aware of formal sources and schooling in locality has a positive impact on the use these sources. On the other hand, schooling has a negative effect on using pawnshops. Availability of banks and savings banks has a positive effect on awareness of those sources but has no effect on their use. Availability of formal sources also has a negative effect on use of some informal sources which suggests that substitution between formal and informal sources exists. Additionally, we find that awareness and use of informal sources is positively affected by household size and that women are more likely to use pawnshops.

Moreover, informal credit institutions, particularly friends and family, play an important role for households when they suffer an income shock. This suggests that the informal sector acts as a complement of the formal sector due to its characteristics, such as flexibility and quick approval of loans, that might be valuable per se in certain situations such as emergencies.

References

- [1] Andrews, R., and Srinivasan, T.C., 1995. Studying Consideration Effects in Empirical Choice Models Using Scanner Panel Data. Journal of Marketing Research. Vol. XXXII, pp. 30-41.
- [2] Anderson, Siwan and Baland, Jean-Marie, 2002. The Economics Of Roscas And Intrahousehold Resource Allocation. The Quarterly Journal of Economics, MIT Press, vol. 117(3), pages 963-995.
- [3] Beck, Thorsten, Demirguc-Kunt, Asli and Levine, Ross, 2004. Finance, Inequality and Poverty: Cross-Country Evidence. World Bank Policy Research Working Paper 3338.
- [4] Beck, Thorsten, Demirgüç-Kunt, Asli and Martinez Peria, Maria Soledad, 2008. Banking Services for Everyone? Barriers to Bank Access and Use around the World. World Bank Economic Review, Oxford University Press, Vol. 22(3), pages 397-430.
- [5] Beck, Thorsten and de la Torre, Augusto, 2007. Analytics of access to financial services. Open Access publications from Tilburg University.
- [6] Bell, C, 1990. Interactions Between Institutional and Informal Credit Agencies in Rural India. The Wold Bank Economic Review 4: 297-327.
- [7] Bendig, M., Giesbert, I., and Stainer S., 2009. Savings, Credit and Insurance: Household Demand for Formal Financial Services in Rural Ghana. German Institute of Global and Area Studies, Working Paper.
- [8] Besley, Timothy, 1995. Nonmarket Institutions for Credit and Risk Sharing in Low-Income Countries. The Journal of Economic Perspectives, Vol. 9, No. 3 (Summer, 1995), pp. 115-127.
- [9] Besley, Timothy and Levenson, Alec R., 1996. The anatomy of an informal financial market: Rosca participation in Taiwan. Journal of Development Economics, Vol. 51 pages 45-68.
- [10] Braverman, A, and Stiglitz, J.E., 1989. Credit rationing, tenancy, productivity, and the dynamics of inequality. In: Bardhan, P. (Ed.), The Economic Theory of Agrarian Institutions. Oxford Univ. Press, New York.
- [11] Chakravarti, Amitav and Janiszewski, Chris, 2003. The Influence of Macro-Level Motives on Consideration Set Composition in Novel Purchase Situations. Journal of Consumer Research, Vol. 30, September.

- [12] Claessens, S., 2005. Access to Financial Services: A review of the Issues and Public Policy Objectives. World Bank WPS3589.
- [13] Claessens, S., 2006. Access to Financial Services: A Review of the Issues and Public Policy Objectives. The World Bank Research Observer, vol. 21, no. 2.
- [14] Clarke, George, Xu, Lixin Colin and Zou, Heng-fu, 2003. Finance and Income Inequality: Test of Alternative Theories. Policy Research Working Paper 2984. World Bank, Washington, D.C.
- [15] Conning, Jonathan and Udry, Christopher, 2007. Rural Financial Markets in Developing Countries. Handbook of Agricultural Economics, Elsevier.
- [16] Dehejia, Rajeev and Lleras-Muney, Adriana, 2007. Financial Development and Pathways of Growth: State Branching and Deposit Insurance Laws in the United States, from 1900 to 1940. Journal of Law and Economics, University of Chicago Press, vol. 50, pages 239-272.
- [17] Demirgüç-Kunt, Asli and Levine, Ross, 2008. Finance and Economic Opportunity. World Bank Policy Research Working Paper 4468, Washington, D.C.
- [18] Eswaran, M. and Kotwal, A., 1989. Credit and agrarian class structure. In: Bardhan, P. (Ed.), The Economic Theory of Agrarian Institutions. Oxford Univ. Press, New York.
- [19] Greene, W., 2006. A general approach to incorporating Selectivity in a model. Working Paper EC-06-10, Department of Economics, Stern School of Business, New York University.
- [20] Guirkinger, Catherine, 2008. Understanding the Coexistence of Formal and Informal Credit Markets in Piura, Peru. World Development. 36(8), 1436-1452.
- [21] Haan, Peter and Uhlendorff, Arne, 2006. Estimation of multinomial logit models with unobserved heterogeneity using maximum simulated likelihood. The Stata Journal, 6, Number 2, pp. 229-245.
- [22] Hernández-Trillo, Fausto, Pagán, José A. and Paxton, Julia, 2005. Start-up Capital, Microenterprises and Technical Efficiency in Mexico. Review of Development Economics Volume 9, Issue 3, pages 434–447, August 2005.
- [23] Honohan, Patrick. 2004. Financial Sector Policy and the Poor: Selected Findings and Issues. World Bank Working Paper 43. Washington, D.C.

- [24] Kiiza, Bernabas and Pederson, Glenn, 2002. Household Financial Savings Mobilisation: Empirical Evidence from Uganda. Journal of African Economies, Volume 10, Number 4, Pp. 390–409.
- [25] King, Robert and Levine, Ross, 1993. Finance, Entrepreneurship and Growth: Theory and Evidence. Journal of Monetary Economics 32, 513-42.
- [26] Kochar, A., 1997. An Empirical Investigation of Rationing Constraints in Rural Credit Markets in India. Journal of Development Economics 53:339–71.
- [27] Levine, Ross, 2005. Finance and Growth: Theory and Evidence. Handbook of Economic Growth, in: Philippe Aghion and Steven Durlauf (ed.), Handbook of Economic Growth, edition 1, volume 1, chapter 12, pages 865-934 Elsevier.
- [28] Okten, Cagla and Osili, Una Okonkwo, 2004. Social Networks and Credit Access in Indonesia. World Development, Vol. 32, No. 7, pp. 1225–1246.
- [29] Pal, S., 2002. Household sectorial choice and effective demand for rural credit in India. Applied Economics, 14, 1743-1755.
- [30] Peachey, Steven, and Roe, Alan, 2004. Access to Finance, A Study for the World Savings Banks Institute. Oxford Policy Management.
- [31] Pearlman, Sarah, 2010. Flexibility Matters: Do More Rigid Loan Contracts Reduce Demand for Microfinance? background paper for CAF's 2011 Reporte de Economia y Desarrollo (RED).
- [32] PROFECO, 2010. La ruta del pignorante inteligente. (Can be seen in: http://www.profeco.gob.mx/profeco/c_emp/R_pignorante.pdf)
- [33] Rajan, Rhaguram and Zingales, Luigi, 1998. Financial Dependence and Growth. American Economic Review 88, 559-587.
- [34] Tang, S., Guan, Z., and Jin S., 2010. Formal and Informal Credit Market and Rural Credit Demand in China. Paper selected for presentation at the Agricultural and Applied Economics Association 2010.
- [35] Terza, Joseph V., 2002. Alcohol Abuse and Employment: A Second Look. Journal of Applied Econometrics, Vol. 17, No. 4, 2002, pp. 393-404.

- [36] Toby, Ronald P., 1991. Both a Borrower and a Lender Be. From Village Moneylender to Rural Banker in the Tempo Era. Monumenta Nipponica, Vol. 46, No. 4 (Winter, 1991), pp. 483-512.
- [37] Varghese, Adel, 2005. Bank-moneylender linkage as an alternative to bank competition in rural credit markets. Oxford Economic Papaper, Volume 57, Issue 2, pp. 315-335.
- [38] World Bank, 2005. Core Concepts and Definitions for Measuring Financial Access in Households Surveys. Report of a Working Group on Harmonizing Households Surveys on Access to Finance. Office of the Vice President, Financial Sector, Washington, D.C.
- [39] Wydick, Bruce, Hayes, Harmony Karp and Kempf, Sarah Hilliker, 2011. Social Networks, Neighborhood Effects, and Credit Access: Evidence from Rural Guatemala. World Development, Vol. 39, No. 6, pp. 974–982.
- [40] Zeller, M.,1994. Determinants of Credit Rationing: A Study of Informal Lenders and Formal Credit Groups in Madagascar. World Development, Vol. 22, No. 12, pp. 1895-1907.

Table 1: Access description

Panel A: Awareness

Number of	Households	Percentage	
sources known	Households	1 creentage	
0	3784	46.9%	
1	2875	35.6%	
2	853	10.6%	
3	324	4.0%	
4	166	2.1%	
5	54	0.7%	
6	9	0.1%	
Total	8,065	100.0%	

Source	Aware of the source	Aware of that only source	+Framily or friends	+Any other
Bank	12.8%	25.6%	48.7%	25.7%
Savings bank	9.7%	30.4%	47.9%	21.7%
Government credit	1.9%	29.8%	52.3%	17.9%
Money lender	10.6%	29.0%	54.5%	16.5%
Pawnshop	8.9%	24.9%	50.3%	24.8%
Family and Friend	35.7%	63.0%	-	37.0%

Panel B: Use

	Ask for loan (use)	Knows credit source (awareness)	Use conditional on awareness
Bank	0.7%	12.8%	5.4%
Savings bank	2.3%	9.7%	23.9%
Government credit	0.6%	1.9%	31.8%
Moneylender	1.5%	10.6%	14.0%
Pawnshop	1.0%	8.9%	11.7%
Family or friends	10.1%	35.7%	28.4%
Total	16.3%		

Observations are number of households in the sample.

Source: Mexican Family Life Survey 2002

Table 2: Summary Statistic

Table 2: Summary Statistic	Mean
Household controls	
Household size	4.25
	(2.04)
Households with male head (proportion)	77.9%
Household head's age	43.79
	(15.80)
Head of house formally employed (proportion)	13.8%
Head of house informally employed (proportion)	51.7%
Average schooling of head and spouse (years)	8.11
	(3.02)
Anual household income (pesos)	33,715
	(64,999)
Negative shock (proportion)	29.9%
Formal savings (proportion)	17.2%
Does not own a house (proportion)	26.3%
Temporary migrant (proportion)	9.5%
Locality controls	
Urban (proportion)	59.0%
Average household annual income in locality (pesos)	39,047
	(19,782)
Average schooling in locality (years)	7.88
	(1.37)
Bank in locality (proportion)	35.2%
Savings bank in locality (proportion)	29.5%
Government credit in locality (proportion)	60.0%
Pawnshop in locality (proportion)	21.7%
Instruments	
Previous serious accident or health problem	16.1%
Previous natural disaster in locality	46.9%
Public transport in locality	68.9%
Change in infrastructure in locality	88.6%

 $[*]Standard\ deviation\ in\ parenthesis.\ Data\ presented\ using\ weights.$

Source: Mexican Family Life Survey 2002

Table 3: Credit sources in a locality

Number of	Number of	Percentage	Number of	Percentage
credit sources	localities	localities	households	households
0	59	39.1%	2680	33.2%
1	48	31.8%	2198	27.2%
2	22	14.6%	1372	17.0%
3	15	9.9%	1209	15.0%
4	7	4.6%	612	7.6%
Total	151	100.0%	8071	100.0%

Source: Mexican Family Life Survey 2002

Table 4: Panel A - Awareness correcting for selection

Probability that a source of credit is considered in a choice set using a normal distribution

Dependent variable is awareness of sources of credit

Household controls	Bank	Savings Bank	Government credit	Money Lender	Pawnshop	Family and Frier
Household size	0.0317	0.0134	0.0118	0.0420**	0.0495*	0.0704***
Household Size	(0.0196)	(0.0172)	(0.0413)	(0.0205)	(0.0275)	(0.0164)
Household head's sex	0.0285	0.0541	0.00424	0.378**	0.207	0.198*
	(0.108)	(0.125)	(0.227)	(0.149)	(0.139)	(0.101)
Household head's age	0.0104**	-0.00499	0.000222	0.00449	0.00343	0.00329
	(0.00519)	(0.00522)	(0.00831)	(0.00548)	(0.00498)	(0.00414)
Household head's age squared	6.86e-05	0.000175***	4.61e-05	7.18e-06	3.13e-05	-1.10e-05
•	(5.01e-05)	(4.77e-05)	(9.54e-05)	(5.84e-05)	(5.34e-05)	(5.23e-05)
Head of house formally employed	0.146	0.586***	-0.184	0.0855	0.132	-0.162
	(0.179)	(0.175)	(0.254)	(0.184)	(0.206)	(0.135)
Head of house informally employed	0.103	0.230*	-0.492***	0.245	-0.0696	-0.0377
	(0.138)	(0.139)	(0.171)	(0.152)	(0.183)	(0.107)
Average schooling of head and spouse	0.0467***	0.0321*	-0.0205	0.0148	0.0176	0.0187
	(0.0121)	(0.0171)	(0.0240)	(0.0118)	(0.0141)	(0.0133)
Anual income (log)	0.186***	0.173***	0.0948*	0.0369	0.0626	0.0672**
	(0.0528)	(0.0477)	(0.0552)	(0.0477)	(0.0489)	(0.0271)
Negative shock	0.0625	0.0852	0.246*	0.111	0.0846	0.142**
	(0.0714)	(0.0851)	(0.138)	(0.0768)	(0.0967)	(0.0622)
Locality controls						
Urban	0.167	0.0914	-0.00629	0.0154	0.260	0.0285
	(0.119)	(0.195)	(0.215)	(0.178)	(0.174)	(0.115)
Average income in locality (log)	0.0975	-0.0823	-0.0896	-0.195	0.220	-0.114
	(0.155)	(0.178)	(0.200)	(0.182)	(0.219)	(0.102)
Average schooling in locality	-0.00212	-0.155*	0.117	-0.0137	0.0301	-0.0636
	(0.0524)	(0.0840)	(0.0794)	(0.0696)	(0.0668)	(0.0395)
Bank in locality	0.243*	0.101	0.236	0.118	0.235	0.116
	(0.132)	(0.227)	(0.184)	(0.140)	(0.228)	(0.0772)
Savings bank in locality	0.209**	0.494***	0.188	0.361**	0.0445	-0.0192
Government credit in	(0.0960)	(0.163) 0.00703	(0.187) 0.195	(0.154) 0.0212	(0.150)	(0.0997) 0.0691
locality	-0.117	0.00703	0.193	0.0212	-0.106	0.0091
	(0.0930)	(0.141)	(0.172)	(0.139)	(0.119)	(0.0797)
Pawnshop in locality	-0.153	0.169	-0.513**	-0.131	0.0141	0.128
	(0.150)	(0.243)	(0.205)	(0.158)	(0.265)	(0.105)
Instruments						
Natural disaster in locality	-0.0243	-0.153	0.346**	-0.323***	0.0314	0.0248
D	(0.106)	(0.147)	(0.157)	(0.111)	(0.210)	(0.0747)
Previous serious accident or health problem	-0.0700	0.195**	0.193	-0.0363	0.00383	0.144*
TO 1 12	(0.0897)	(0.0919)	(0.153)	(0.0941)	(0.122)	(0.0859)
Public transport in locality	-0.0555	0.166	0.128	0.193	0.225*	0.0299
Change in infrastructure in	(0.123)	(0.162) -0.0752	(0.236)	(0.217) -0.208	(0.131)	(0.0989) 0.165*
locality	(0.114)	(0.167)	(0.199)	(0.136)	(0.154)	(0.0992)
Constant	-5.134*** (1.326)	-2.123 (1.742)	-3.122* (1.753)	-0.494 (1.580)	-5.593** (2.266)	-0.347 (0.906)

Note: *** significant at 0.1% level **significant at 1% level * significant at 5% level. Robust errors with clusters by locality in parenthesis, there are 129 localities in the sample. The number of observations is smaller from the number of households in the survey due to missing values in some variables. Analytical weights used.

Table 4: Panel B - Use of credit conditional on awareness correcting for selection

Multinomial logit with sample selection and consideration set

Dependent variable is use of sources of credit

Dependent variable is use of			Government credit	Money Lender	Pawnshop	Family and Friend
Household controls					r	
Household size	0.103	-0.0414	0.386	0.195**	0.269*	0.0618
	(0.178)	(0.0859)	(0.320)	(0.0942)	(0.142)	(0.0469)
Household head's sex	-0.589	0.996	1.594	0.945	-1.388*	0.0697
	(0.696)	(0.611)	(1.165)	(0.756)	(0.827)	(0.250)
Household head's age	-0.145***	-0.0649**	0.305*	0.0629**	-0.00983	-0.00648
	(0.0274)	(0.0270)	(0.177)	(0.0284)	(0.0209)	(0.0184)
Household head's age squared	0.00116***	1.94e-05	-0.00315*	-0.000754***	-0.000178	-0.000112
•	(0.000413)	(0.000242)	(0.00180)	(0.000221)	(0.000349)	(0.000196)
Head of house formally employed	1.743*	-0.405	-3.411**	-0.179	-1.159	0.212
	(1.020)	(0.790)	(1.414)	(0.717)	(0.845)	(0.266)
Head of house informally employed	1.323	-0.585	1.023	-0.605	-0.661	-0.370
	(0.835)	(0.735)	(1.288)	(0.508)	(0.979)	(0.251)
Average schooling of head and spouse	-0.0326	-0.146**	0.202	0.108	-0.106*	0.0153
	(0.147)	(0.0739)	(0.249)	(0.0845)	(0.0626)	(0.0356)
Anual income (log)	0.717***	-0.0626	-0.869	-0.443*	-0.0767	-0.229***
	(0.273)	(0.217)	(0.791)	(0.231)	(0.368)	(0.0849)
Negative shock	0.670	0.129	-0.141	0.0468	-0.158	0.547***
	(0.656)	(0.339)	(0.886)	(0.361)	(0.679)	(0.144)
Locality controls						
Urban	-1.419	-0.338	0.474	0.978**	-0.0190	0.597**
	(0.879)	(0.715)	(2.555)	(0.469)	(1.034)	(0.248)
Average income in locality (log)	-2.437***	-1.256*	-0.792	-0.604	-2.478**	-1.044***
	(0.716)	(0.674)	(3.567)	(0.399)	(1.202)	(0.208)
Average schooling in locality	0.725***	0.117	2.196**	-0.0165	0.0194	0.0277
	(0.263)	(0.203)	(0.937)	(0.249)	(0.544)	(0.0922)
Bank in locality	0.119	0.786*	-1.312	-1.093**	0.848	0.144
	(0.685)	(0.459)	(1.263)	(0.425)	(1.055)	(0.239)
Savings bank in locality	-0.789	-0.843	3.500	-0.486	-0.413	-0.637***
	(0.738)	(0.585)	(2.359)	(0.383)	(0.883)	(0.236)
Government credit in locality	0.868	-0.579	1.501	0.138	0.0517	0.188
	(0.730)	(0.503)	(1.397)	(0.304)	(0.727)	(0.184)
Pawnshop in locality	0.873	1.389**	-3.148	0.131	-0.447	0.669**
	(0.736)	(0.593)	(2.045)	(0.690)	(0.794)	(0.263)
Constant	13.01*	17.12**	-13.07	6.372*	28.46**	11.94***
	(7.069)	(6.858)	(29.07)	(3.592)	(12.92)	(2.285)
Sigma	1.316	1.685***	0.964	1.073**	2.115**	0.514***
~	(0.924)	(0.419)	(0.748)	(0.460)	(1.001)	(0.129)
Rho	-0.862***	-0.487***	-0.413***	-0.502***	-0.617***	-0.293***
	(0.0590)	(0.0786)	(0.101)	(0.0731)	(0.101)	(0.0491)
Observations	3,683	3,683	3,683	3,683	3,683	3,683
Observations	2,002	2,002	2,003	2,002	2,002	2,003

Note: *** significant at 0.1% level **significant at 1% level * significant at 5% level. Robust errors with clusters by locality in parenthesis, there are 129 localities in the sample. The number of observations is smaller from the number of households in the survey due to missing values in some variables. Analytical weights used.

Table 5: Panel A - Coefficient interpretation, marginal effects on awareness correcting for selection Probability that a source of credit is considered in a choice set using a normal distribution

Dependent variable is awareness of sources of credit

Dependent variable is aware	Bank		Government credit	Manari I andan	Darringhan	Family and Eriand
Household controls	Dalik	Savings Dank	Government credit	Wioney Lender	Pawiisiiop	ranning and rhend
Household size	0.0063	0.0023	0.0004	0.0076***	0.0066*	0.0269***
	0.0063	0.0023	-0.004	0.0076****	0.0203	0.02693*
Household head's sex						
Household head's age	0.0021*	-0.0008	0.00001	0.0008	0.0005	0.0013
Household head's age squared	0.00002	0.00003***	0.000002	0.000002	0.000005	-0.000004
Head of house formally employed	0.0402	0.1032***	-0.0045	0.0211	0.022	-0.0581
Head of house informally employed	0.0328	0.0445*	-0.0139***	0.0515	-0.0031	-0.0092
Average schooling of head and spouse	0.0095***	0.0055*	-0.0006	0.0028	0.0025	0.0073
Anual income (log)	0.0374***	0.0286***	0.003*	0.0063	0.008	0.0254**
Negative shock	0.0122	0.014	0.0087*	0.0199	0.0109	0.0541**
Locality controls						
Urban	0.0296	0.0139	-0.0005	0.0007	0.0306	0.0095
Average income in	0.01.11	0.0161	0.0025	0.0272	0.02.62	0.0445
locality (log)	0.0141	-0.0161	-0.0035	-0.0373	0.0262	-0.0447
Average schooling in locality	-0.0001	-0.0256*	0.0038	-0.0021	0.0043	-0.0242
Bank in locality	0.05*	0.0164	0.0078	0.0206	0.032	0.0437
Savings bank in locality	0.0407**	0.0928***	0.0061	0.0699***	0.005	-0.0084
Government credit in locality	-0.0237	0.0008	0.0059	0.0034	-0.0145	0.0265
Pawnshop in locality	-0.0247	0.0327	-0.0115**	-0.0201	0.0041	0.0517
Instruments						
Natural disaster in locality	-0.0071	-0.0259	0.0113**	-0.0576***	0.003	0.0088
Previous serious accident or health problem	-0.0114	0.0373***	0.0073	-0.0048	0.0023	0.0569*
Public transport in locality	-0.0051	0.0299	0.0048	0.0371	0.0321*	0.0142
Change in infrastructure in locality	-0.081***	-0.0161	-0.0086	-0.0446	-0.0308	0.0591*

Note: Marginal effects are calculated at the mean. *** significant at 0.1% level **significant at 1% level * significant at 5% level.

Table 5: Panel B - Coefficient interpretation, odds ratios of use conditional on awareness correcting for selection. From multinomial logit with sample selection and consideration sets

Dependent variable is use of each source

Dependent variable is use of	Dependent variable is use of each source									
	Bank	Savings Bank	Government credit	Money Lender	Pawnshop	Family and Friend				
Household controls										
Household size	1.11	0.96	1.47	1.22**	1.30*	1.06				
Household head's sex	0.55	2.71	4.92	2.57	0.25*	1.07				
Household head's age	0.87***	0.94**	1.36*	1.06**	0.99	0.99				
Household head's age squared	1.001***	1.00	0.99*	0.99***	1.00	1.00				
Head of house formally employed	5.71*	0.67	0.03**	0.84	0.31	1.24				
Head of house informally employed	3.75	0.56	2.78	0.55	0.52	0.69				
Average schooling of head and spouse	0.97	0.86**	1.22	1.11	0.90*	1.02				
Anual income (log)	2.05***	0.94	0.42	0.64	0.93	0.80***				
Negative shock	1.95	1.14	0.87	1.05	0.85	1.73***				
Locality controls										
Urban	0.24	0.71	1.61	2.66**	0.98	1.82**				
Average income in locality (log)	0.09***	0.28*	0.45	0.55	0.08**	0.35***				
Average schooling in locality	2.06***	1.12	8.99**	0.98	1.02	1.03				
Bank in locality	1.13	2.19*	0.27	0.91**	2.33	1.15				
Savings bank in locality	0.45	0.43	33.12	0.62	0.66	0.53***				
Government credit in locality	2.38	0.56	4.49	1.15	1.05	1.21				
Pawnshop in locality	2.39	4.01**	0.04	1.14	0.64	1.95**				

Note: Odds ratios must be interpreted with respect to the option not asking. An odds ratio of 1.05 means that the variable makes it 5% more likely to ask for a credit at that source than not ask for a credit.

Table 6: Multinomial LogitDependent variable is use of each source

Dependent variable is use of		Coving- D 1	Corroman 1'	Monayitana	Darry1	Equility on J. Enis. 1
Household controls	Bank	Savings Bank	Government credit	Money Lender	Pawnshop	ramily and Friend
Household size	0.122	0.0305	0.141	0.144**	0.311***	0.127***
TIOUSCHOIG SIZE	(0.138)	(0.0459)	(0.157)	(0.0706)	(0.119)	(0.0315)
Household head's sex	-0.585	0.962**	-0.0707	2.205***	-0.507	0.356
Household head's sex	(0.563)	(0.381)	(0.557)	(0.629)	(0.479)	(0.229)
Household head's age	-0.0737***	-0.0440***	0.0720*	0.0903**	0.00375	0.00455
Household hours age	(0.0168)	(0.0135)	(0.0437)	(0.0393)	(0.0196)	(0.0126)
Household head's age squared	0.000918***		-0.000862***	-0.000916**	-0.000220	-0.000137
•	(0.000188)	(0.000116)	(0.000287)	(0.000374)	(0.000325)	(0.000125)
Head of house formally employed	1.349*	0.437	-2.168**	-0.0302	-0.888	-0.157
	(0.787)	(0.428)	(0.926)	(0.604)	(0.632)	(0.272)
Head of house informally employed	0.793	-0.440	-1.197**	-0.0978	-0.822	-0.304
	(0.494)	(0.419)	(0.471)	(0.492)	(0.583)	(0.238)
Average schooling of head and spouse	-0.00474	-0.00859	0.0195	0.0739	-0.0700*	0.0417
	(0.0666)	(0.0447)	(0.0927)	(0.0632)	(0.0409)	(0.0295)
Anual income (log)	1.149***	0.359***	0.0487	-0.295**	0.0658	-0.0874
	(0.175)	(0.136)	(0.128)	(0.129)	(0.176)	(0.0608)
Negative shock	0.632	0.386	0.218	0.370	-0.226	0.547***
Locality controls	(0.599)	(0.272)	(0.438)	(0.257)	(0.490)	(0.133)
Urban	-0.196	0.299	1.393	0.830*	1.173	0.483*
Croan	(0.676)	(0.698)	(1.147)	(0.491)	(0.722)	(0.258)
Average income in locality						
(log)	-1.509**	-0.872	-1.509**	-0.701*	-0.656	-0.794***
	(0.593)	(0.599)	(0.659)	(0.400)	(0.541)	(0.174)
Average schooling in locality	0.532***	-0.233	0.397*	0.0744	0.268	-0.0467
	(0.170)	(0.250)	(0.216)	(0.227)	(0.206)	(0.0955)
Bank in locality	0.0813	0.804*	0.722	-0.814*	0.306	0.203
	(0.540)	(0.465)	(0.968)	(0.442)	(0.561)	(0.227)
Savings bank in locality	-0.558	0.813*	0.720	0.386	0.279	-0.395
	(0.625)	(0.439)	(0.705)	(0.420)	(0.514)	(0.244)
Government credit in locality	0.392	-0.290	1.360*	0.0991	0.0907	0.203
	(0.613)	(0.446)	(0.725)	(0.385)	(0.496)	(0.177)
Pawnshop in locality	0.486	0.741	-1.842**	-0.559	-0.177	0.476*
	(0.644)	(0.586)	(0.863)	(0.468)	(0.642)	(0.270)
Constant	-6.100	3.273	3.920	-0.0252	-1.213	5.962***
	(5.699)	(4.615)	(7.920)	(4.358)	(4.699)	(1.562)
Ohaamatiana	2.040	2.040	2.040	2.040	2.040	2.040
Observations Note: *** significant at 0.1% lev	3,840	3,840	3,840	3,840	3,840	3,840

Note: *** significant at 0.1% level **significant at 1% level * significant at 5% level. Robust errors with clusters by locality in parenthesis, there are 129 localities in the sample. The number of observations is smaller from the number of households in the survey due to missing values in some variables. The results should be interpreted with reference to not using any type of credit. Analytical weights used.

Table 7: Panel A - AwarenessProbability that a source of credit is considered in a choice set using a normal distribution Dependent variable is awareness of sources of credit

Dependent variable is awaren	Bank		Government credit	Money Lender	Pawnshop	Family and Friend
Household controls				<u>, </u>		,
Household size	0.0283	0.00787	0.0140	0.0442**	0.0437	0.0743***
	(0.0196)	(0.0166)	(0.0350)	(0.0176)	(0.0303)	(0.0161)
Household head's sex	0.0395	0.0828	0.0609	0.345**	0.223	0.213**
	(0.112)	(0.120)	(0.207)	(0.144)	(0.140)	(0.0935)
Household head's age	0.0116**	-0.00623	0.000910	0.00395	0.00392	0.00374
_	(0.00555)	(0.00493)	(0.00843)	(0.00551)	(0.00522)	(0.00434)
Household head's age squared	6.63e-05	0.000182***	3.71e-05	2.37e-05	2.98e-05	-8.77e-06
	(5.40e-05)	(4.66e-05)	(9.85e-05)	(5.70e-05)	(5.73e-05)	(5.25e-05)
Head of house formally employed	0.152	0.567***	-0.216	0.136	0.114	-0.174
	(0.193)	(0.170)	(0.271)	(0.172)	(0.218)	(0.128)
Head of house informally employed	0.110	0.234*	-0.500***	0.270*	-0.0824	-0.0510
	(0.152)	(0.131)	(0.184)	(0.149)	(0.191)	(0.106)
Average schooling of head and spouse	0.0496***	0.0343**	-0.0242	0.0168	0.0173	0.0188
	(0.0130)	(0.0167)	(0.0228)	(0.0120)	(0.0143)	(0.0128)
Anual income (log)	0.181***	0.165***	0.103*	0.0369	0.0653	0.0760***
	(0.0612)	(0.0502)	(0.0527)	(0.0512)	(0.0512)	(0.0265)
Negative shock	0.0574	0.0882	0.195	0.0768	0.0382	0.137**
	(0.0804)	(0.0839)	(0.132)	(0.0748)	(0.0931)	(0.0617)
Locality controls						
Urban	0.0978	0.0966	0.0300	0.0517	0.265	0.0786
	(0.139)	(0.199)	(0.259)	(0.161)	(0.178)	(0.111)
Average income in locality (log)	0.162	-0.00341	-0.202	-0.0113	0.232	-0.146
	(0.134)	(0.170)	(0.187)	(0.170)	(0.186)	(0.0963)
Average schooling in locality	-0.0295	-0.132	0.0643	-0.0106	0.0237	-0.0604
	(0.0540)	(0.0804)	(0.0824)	(0.0719)	(0.0670)	(0.0372)
Bank in locality	0.234*	0.0834	0.465**	0.0284	0.313	0.127*
	(0.124)	(0.234)	(0.225)	(0.139)	(0.211)	(0.0748)
Savings bank in locality	0.207**	0.499***	0.157	0.349**	0.0374	-0.0322
	(0.0982)	(0.161)	(0.220)	(0.154)	(0.160)	(0.0988)
Government credit in locality	-0.151	-0.0232	0.203	-0.0143	-0.125	0.107
	(0.0954)	(0.139)	(0.153)	(0.134)	(0.124)	(0.0774)
Pawnshop in locality	-0.185	0.153	-0.391*	-0.188	0.00184	0.140
	(0.149)	(0.228)	(0.209)	(0.171)	(0.224)	(0.102)
Constant	-5.880***	-3.000*	-1.597	-2.607*	-5.681***	-0.0137
	(1.101)	(1.605)	(1.734)	(1.419)	(1.678)	(0.836)
Observations	3,840	3,840	3,840	3,840	3,840	3,840
			· · ·	* -		** *

Note: *** significant at 0.1% level **significant at 1% level * significant at 5% level. Robust errors with clusters by locality in parenthesis, there are 129 localities in the sample. The number of observations is smaller from the number of households in the survey due to missing values in some variables. Analytical weights used.

Table 7: Panel B - Use of credit conditional on awareness

Multinomial logit with consideration set

Dependent variable is use of sources of credit

Dependent variable is use of	Bank		Government credi	t Monay I andar	Dawnshon	Family and Friend
Household controls	Dalik	Savings Dank	Government credi	t Money Lender	rawiisiiop	ranniy and Friend
Household size	0.141	0.0585	0.406	0.165*	0.266**	0.0614
Trousehold size	(0.169)	(0.0613)	(0.264)	(0.0856)	(0.134)	(0.0451)
Household head's sex	-0.451	0.921*	1.039	1.193*	-0.801	0.137
Household head's sex	(0.611)	(0.471)	(0.812)	(0.666)	(0.562)	(0.241)
Hoyaahald baadla aaa	-0.125***	-0.0427**	0.220	0.0656**	-0.0118	-0.00842
Household head's age						
TT	(0.0224)	(0.0182)	(0.160)	(0.0318)	(0.0210)	(0.0159)
Household head's age squared	0.00114***	3.38e-05	-0.00223	-0.000810***	-0.000198	-8.78e-05
	(0.000372)	(0.000181)	(0.00171)	(0.000280)	(0.000381)	(0.000170)
Head of house formally employed	1.527	-0.448	-2.260**	-0.313	-0.794	0.121
	(0.931)	(0.637)	(1.032)	(0.600)	(0.740)	(0.250)
Head of house informally employed	1.048	-0.710	0.925	-0.622	-0.542	-0.337
	(0.730)	(0.595)	(1.219)	(0.466)	(0.774)	(0.231)
Average schooling of head and spouse	0.00503	-0.104**	0.239	0.0929	-0.115**	0.0140
	(0.128)	(0.0511)	(0.200)	(0.0813)	(0.0555)	(0.0323)
Anual income (log)	0.824***	0.107	-0.601	-0.382**	0.00375	-0.243***
	(0.251)	(0.194)	(0.582)	(0.173)	(0.233)	(0.0796)
Negative shock	0.797	0.258	-0.599	0.173	0.00220	0.532***
	(0.614)	(0.249)	(0.903)	(0.311)	(0.540)	(0.134)
Locality controls	, ,			, ,	, ,	, ,
Urban	-1.217*	-0.0328	2.133	0.878**	0.221	0.532**
	(0.698)	(0.599)	(2.544)	(0.414)	(0.744)	(0.240)
Average income in locality (log)	-2.254***	-1.213**	-2.353	-0.685**	-1.989**	-0.971***
	(0.640)	(0.529)	(3.095)	(0.345)	(0.980)	(0.193)
Average schooling in locality	0.689***	0.0622	1.764*	0.0769	0.358	0.0494
	(0.228)	(0.176)	(0.902)	(0.213)	(0.355)	(0.0877)
Bank in locality	0.192	0.685	-1.542	-1.085**	0.180	0.148
	(0.639)	(0.434)	(1.243)	(0.437)	(0.681)	(0.233)
Savings bank in locality	-0.674	-0.432	2.640*	-0.439	0.151	-0.559**
	(0.674)	(0.440)	(1.593)	(0.349)	(0.563)	(0.234)
Government credit in locality	0.657	-0.380	1.670	0.205	0.152	0.153
	(0.662)	(0.421)	(1.046)	(0.301)	(0.548)	(0.176)
Pawnshop in locality	0.909	0.807*	-3.088*	-0.103	-0.397	0.504*
. ,	(0.662)	(0.474)	(1.691)	(0.585)	(0.590)	(0.265)
Constant	7.586	12.39**	4.225	5.376*	17.64**	11.07***
	(5.300)	(4.875)	(23.17)	(3.021)	(8.602)	(2.056)
Observations	3,840	3,840	3,840	3,840	3,840	3,840

Note: *** significant at 0.1% level **significant at 1% level * significant at 5% level. Robust errors with clusters by locality in parenthesis, there are 129 localities in the sample. The number of observations is smaller from the number of households in the survey due to missing values in some variables. Analytical weights used.

Table 8: Awareness and use conditional on awareness correcting for selection

Dependent variable are awareness and use of each source

Dependent variable are awareness and use of each source Formal Informal				
	Awareness Use		Awareness Use	
Household controls				
Household size	0.0287	0.0600	0.0750***	0.205
	(0.0191)	(0.0623)	(0.0154)	(1.009)
Household head's sex	0.0335	0.339	0.262***	-0.628
	(0.0988)	(0.289)	(0.0845)	(3.974)
Household head's age	4.39e-05	-0.0581***	0.00515	0.00239
	(0.00346)	(0.0138)	(0.00457)	(0.101)
Household head's age squared	0.000165***	0.000223	-2.06e-05	-0.000770
-	(3.27e-05)	(0.000184)	(4.56e-05)	(0.00456)
Head of house formally employed	0.284*	-0.110	-0.0101	-0.454
	(0.171)	(0.500)	(0.144)	(3.015)
Head of house informally employed	0.0640	-0.422	0.0303	-1.890
	(0.125)	(0.413)	(0.117)	(10.03)
Average schooling of head and spouse	0.0415***	-0.0575	0.0175	0.0407
	(0.0138)	(0.0393)	(0.0129)	(0.255)
Anual income (log)	0.217***	0.122	0.0419	-1.090
	(0.0411)	(0.164)	(0.0284)	(5.874)
Negative shock	0.110**	0.265	0.110*	1.751
	(0.0559)	(0.263)	(0.0665)	(9.204)
Locality controls				
Urban	0.104	0.0214	0.0593	2.314
	(0.126)	(0.532)	(0.120)	(12.58)
Average income in locality (log)	0.0155	-1.273**	-0.104	-4.170
	(0.145)	(0.514)	(0.119)	(21.89)
Average schooling in locality	-0.0529	0.0999	-0.0538	0.221
	(0.0601)	(0.167)	(0.0483)	(1.254)
Bank in locality	0.249	0.365	0.141	-0.674
	(0.162)	(0.459)	(0.105)	(4.331)
Savings bank in locality	0.339***	0.0423	0.0460	-2.319
	(0.111)	(0.369)	(0.110)	(12.82)
Government credit in locality	-0.0563	0.258	0.00575	0.645
	(0.0928)	(0.360)	(0.0888)	(3.453)
Pawnshop in locality	-0.0573	0.569	0.0915	1.852
	(0.165)	(0.510)	(0.130)	(9.768)
Instruments				
Natural disaster in locality	-0.0237		-0.0916	
	(0.113)		(0.0909)	
Previous serious accident or health problem	0.0873		0.104	
	(0.0815)		(0.0792)	
Public transport in locality	0.0699		0.211	
	(0.129)		(0.130)	
Change in infrastructure in locality	-0.305***		0.0229	
•	(0.102)		(0.0876)	
Constant	-3.771***	11.96***	-0.111	52.97
	(1.302)	(4.471)	(1.047)	(278.9)
Ciama		0.420*		0 522
Sigma		0.439*		8.532
DI.		(0.244)		(48.19)
Rho		-0.397*** (0.0677)		-0.608*** (0.122)
		(0.0077)		(0.122)
Observations	3,683	3,683	3,683	3,683

Note: *** significant at 0.1% level **significant at 1% level * significant at 5% level. Awareness is the probability that a source of credit is considered in a choice set using a normal distribution. Use is a multinomial logit with sample selection and consideration set. Robust errors with clusters by locality in parenthesis, there are 129 localities in the sample. The number of observations is smaller from the number of households in the survey due to missing values in some variables. Use of each source should be interpreted with reference to not using any type of credit. Analytical weights used.